Author's response to reviews

Title: Predictors of serum dioxin levels among adolescent boys in Chapaevsk, Russia: A cross-sectional study.

Authors:

Russ Hauser (rhauser@hohp.harvard.edu)
Paige Williams (paige@jessica.harvard.edu)
Larisa Altshul (laltshul@hsp.harvard.edu)
Susan Korrick (susan.korrick@channing.harvard.edu)
Lynne Peeples (lpeeples@hsp.harvard.edu)
Donald G Patteson Jr (dgp1@cdc.gov)
Wayman E Turner (we1@cdc.gov)
Mary M Lee (LeeM@ummhc.org)
Boris Revich (revich@mail.ecfor.rssi.ru)
Oleg Sergeyev MD (endopsyh@mail.samtel.ru)

Version: 8 Date: 24 May 2005

Author's response to reviews:

Included Tables 2, 4 and 5 in the text of the manuscript and deleted them as additional files, as requested.